Author's response to reviews

Title: The BRAzil MAGnesium (BRAMAG) Trial: A randomized clinical trial of oral magnesium supplementation in pregnancy for the prevention of preterm birth and perinatal and maternal morbidity

Authors:

Joao GB Alves (joaoguilherme@imip.org.br)
Carla A de Araújo (carlaleal_2@hotmail.com)
Isabelle EA Pontes (isabelle_albuquerque@hotmail.com)
Angélica C Guimarães (angel.guimaraes@bol.com.br)
Joel G Ray (rayj@smh.ca)

Version: 2 Date: 15 June 2014

Dear BMC Pregnancy and Childbirth

Re. MS: 1641826773132300 The BRAzil MAGnesium (BRAMAG) Trial: A randomized clinical trial of oral magnesium.

1. Ethical and Funding Approval Documentation # Research ethics board approval letter is attached.

2. The study is peer review funded by the Brazilian National Council for Scientific and Technological Development (CNPq), the Ministry of Health (MoH) and the Bill & Melinda Gates Foundation. The letters are attached.

3. Study status
The protocol represents a study that has not yet started recruiting participants.

4. Related articles
There are not related articles

5. Consort Guidelines
This study adheres to the CONSORT Guidelines for the reporting of randomized clinical trials. This is added at the end of the manuscript text, which is attached.

Figure 1 is also attached, if needed.

Thank you
Joel Ray